Author’s response to reviews

Title: Case report Pregabalin induced visual hallucinations – A rare adverse reaction

Authors:

Georgios Mousailidis (gmous24@hotmail.com)

Basavaraja Papanna (b.papanna@nhs.net)

Andrew Salmon (andrew.salmon1@nhs.net)

Sein Aung (aung.sein@nhs.net)

Qutaiba Al-Hillawi (q_hillawi@yahoo.com)

Version: 1  Date: 13 Aug 2019

Author’s response to reviews:

Response to comments

It is already documented in the manuscript that the patient never used any drugs in her life and never consumed alcohol.

Her food intake and sleep patterns remained the same and did not change during the side effect. We added one sentence on that.

Hallucinations are almost never evident on EEG.

We followed her up for almost one year and she did not report any visual hallucinations. We added one more sentence on that. We did not perform another EEG as it was not necessary.

We added one more sentence comparing our study with the one previous study reporting visual hallucinations. Our study is the second study in the literature reporting visual hallucinations as a side effect.

Kind Regards
Georgios Mousailidis